UCLA UCLA Previously Published Works

Title

Nomadic Tibetan women's reproductive health: findings from cross-sectional surveys with a hard-to-reach population.

Permalink https://escholarship.org/uc/item/6qz0m5tg

Journal Reproductive health, 18(1)

ISSN 1742-4755

Authors

Gipson, Jessica D Moucheraud, Corrina Gyaltsen, Kunchok <u>et al.</u>

Publication Date 2021-03-01

DOI

10.1186/s12978-020-01052-0

Peer reviewed

RESEARCH

Open Access



Nomadic Tibetan women's reproductive health: findings from cross-sectional surveys with a hard-to-reach population

Jessica D. Gipson^{1*}, Corrina Moucheraud¹, Kunchok Gyaltsen (Gangue Jianzan)², Lumo Tsering (Limao Cairang)², Tabashir Z. Nobari³ and Lhusham Gyal (Li Xianjia)²

Abstract

Background: Western China has undergone substantial sociodemographic change, yet little is known about the health status of ethnic minority populations living in these areas.

Methods: We report findings from two cross-sectional surveys conducted with female Tibetan nomads living in rural areas of Western China/Eastern Tibet. We present results of descriptive analyses of data collected from reproductive-aged females who attended community health fairs in 2014 (n = 193) and 2016 (n = 298).

Results: On average, sexual debut preceded marriage among study participants, with fertility near replacement levels (2.7 and 2.1 in 2014 and 2016, respectively). Contraceptive use was common, and dominated by use of IUDs and female sterilization. Although over three-quarters (76%) of 2016 survey participants reported ever having at least one sexually transmitted infection (STI) symptom, there was low awareness of STIs (59%) and action to prevent STIs (21%). Younger women (<40) were more likely to report having had had an STI symptom, as compared to older women (84% versus 71%; p < 0.05).

Conclusions: We demonstrate feasibility of collecting data with this hard-to-reach population. Reporting of STI symptoms warrants further investigation to identify and address health conditions in this population of Tibetan nomadic women, especially amidst broader social and contextual changes that may affect the Tibetan population.

Plain English Summary

Western China has undergone substantial sociodemographic change, yet little is known about the health status of ethnic minority populations living in these areas. We report findings from two cross-sectional surveys conducted with female Tibetan nomads living in rural areas of Western China/Eastern Tibet. We present results of descriptive analyses of data collected from reproductive-aged females who attended community health fairs in 2014 (n = 193) and 2016 (n = 298). On average, sexual debut preceded marriage among study participants, with fertility near replacement levels (2.7 and 2.1 in 2014 and 2016, respectively). Contraceptive use was common, and dominated by use of IUDs and female sterilization. Although over three-quarters (76%) of 2016 survey participants reported ever having at least one sexually transmitted infection (STI) symptom, there was low awareness of STIs (59%) and action to prevent STIs (21%). Younger women (less than 40 years old) were more likely to report having had had an STI symptom, as compared to

*Correspondence: jgipson@ucla.edu

¹ Fielding School of Public Health, University of California, 650 Charles E.

Young Drive South, Los Angeles, CA 90095, USA

Full list of author information is available at the end of the article



© The Author(s) 2021. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

women over 40 years old (84% versus 71%; p < 0.05). We demonstrate feasibility of collecting data with this hard-toreach population. Reporting of STI symptoms warrants further investigation to identify and address health conditions in this population of Tibetan nomadic women, especially amidst broader social and contextual changes that may affect the Tibetan population.

Keywords: China, Tibetan, Nomad, Reproductive health

Introduction

Western China has historically been home to a large number of ethnic minority residents, but has undergone substantial economic and social change in the past two decades [1, 2]. Government initiatives have prompted economic development in Western China, including the expansion of private industry, leading to an influx of Han (ethnic majority) Chinese into the more rural areas of China and the Tibetan plateau [1, 2]. In addition to these development efforts, policies that promote the settlement of Tibetan populations into urban areas have substantially altered the cultural context of Western China. Despite these myriad changes, these provinces remain some of the poorest in China [3]. Little is known about how ethnic minority populations have fared amidst these sociodemographic shifts [4].

Previous analyses document persistent disparities in health outcomes across Chinese provinces (e.g., maternal mortality; child health) [5, 6], yet there have been fewer investigations of ethnic minority populations in these more-disadvantaged areas [7]. Very little is known about nomadic Tibetan populations, likely due to their combined identity as an ethnic minority group, and as a mobile, difficult-to-reach population. While some evidence has shown health advantages among nomadic populations (e.g., biological adaptations to higher altitudes) [8, 9], other studies indicate that rural ethnic minority populations are more vulnerable to health issues given their distance from health facilities, extreme climate and terrain, linguistic and cultural barriers, and challenging lifestyle and environmental factors (e.g., availability of food and water) [4, 10, 11]. Ongoing political and ethnic tensions may also impede health care-seeking behavior and affect the quality of care provided for Tibetan nomads. Ethnic minority populations may face unique barriers, including communication with health professionals at Chinese health care facilities and discriminatory treatment [12]. Evidence suggests that these nomadic groups may even experience differential treatment by other Tibetans [13]. Despite the challenges, however, a recent analysis of maternal health behaviors among Tibetans in Qinghai Province found that utilization of antenatal care, institutional delivery, and/ or skilled birth attendance has significantly increased over time, from 10% of births prior to the year 2000, to approximately 50% between 2000 and 2014 [14]. Moreover, women were more likely to use these services if they participated in the New Cooperative Medical Scheme, the national health insurance plan available to Chinese citizens [15].

Among the few studies that have examined the sexual and reproductive health status of Tibetan populations [5, 16-19], a focus on nomadic populations is rare. Although Tibetan and other minority populations in China were granted some leniency to China's One-Child Policy (implemented by the Chinese Government in the late 1970s and early 1980s to limit population growth by promoting one child per family), the 2000 census indicated below-replacement fertility levels in the Tibetan autonomous prefectures and somewhat higher in the two most remote nomadic prefectures (Qinghai tabulation 2002: Table L.6-1 by Fischer, 2008). In-depth examinations of fertility and family planning practices in Tibetan communities underscore the sensitivity of these topics, given historical evidence of coercive practices and challenges in reconciling official family planning policy with Tibetan beliefs and practices [1, 19]. However, ethnographic studies by Schrempf [13] and Craig [20] have explored these topics-and describe the sociocultural significance of motherhood and fertility for Tibetan women, including the importance of continuing the family lineage, maintaining karma, and ensuring a woman's status in her partnership and her community [13]. Additionally, contraceptive methods and their documented side effects may run counter to Tibetan cultural schemas surrounding the importance of hard work and strength among Tibetan women, and the extent to which women are knowledgeable about how contraceptives work in the body especially given low education levels, awareness of methods, and discrepancies between Western and Tibetan health beliefs [13].

Examinations of sexual health—including sexually transmitted infections—are extremely limited in ethnic minority populations, and particularly among nomadic populations. National estimates suggest that the prevalence of sexually transmitted infections (STIs) has increased in recent decades [21], and some studies have found relatively higher risks for ethnic minority populations versus Han groups [22–24].

This project sought to document the reproductive health status of an ethnic minority population-Tibetan nomads-living in Qinghai province in Western China. Qinghai Province is part of the traditional Tibetan region of Amdo and has the third largest Tibetan population outside of the Tibetan Autonomous Region and Sichuan (Fischer 2008). Qinghai is consistently ranked below the China average in per capita income and GDP [3]. In 2014 and 2016, a team of public health and medical practitioners collected data from Tibetan nomads living in two areas of Western China to assess their reproductive health status. A separate paper by Moucheraud et al. [14] examined maternal health care behaviors using the 2014 data. Following on those findings, the aim of this paper is to integrate the 2014 and 2016 data to report on the reproductive health status and behaviors of Tibetan nomadic women across these two data collection efforts in this remote, ethnic minority population.

Materials and methods

The two surveys—Site A (2014) and Site B (2016)—were conducted in rural areas of Western China/Eastern Tibet by faculty and students from the Tso-Ngon (Qinghai) University Tibetan Medical College (TUTMC) in Tso-ngon, Tibet/Qinghai, China and (See map; Fig. 1).

Due to the difficulty of collecting data from this hardto-reach, rural and nomadic population, data were collected during 2–3 day community health fairs in two different Tibetan prefectures: Site A in 2014 and Site B in 2016 [14]. (We subsequently refer to these sites as Site A and Site B). The health fairs were advertised via the County Health Department and Women's Union, as well as by Tibetan village leaders and village doctors in the area. The health fairs were open to anyone who attended and drew approximately 1600 people per fair.

During these health fairs, male and female participants were offered free medical consultation from Tibetan and Chinese health professionals and health education sessions; however, only adult women were asked to participate in the health survey. In total, 200 women at Site A and 306 women at Site B consented to participate in the survey.

All data collectors for the two surveys were Tibetan medical students or professionals who were trained on the administration of the survey, including pretesting the surveys with Tibetan students and refining the survey instruments prior to implementation in the field. The two surveys were developed by a team of Tibetan-US researchers to include questions about respondents' birth histories, health care seeking, health beliefs, and general



sociodemographic characteristics, including selected questions from the Demographic Health Survey modules [25]. The surveys were conducted face-to-face in the local language (Amdo Tibetan).

The final analytic samples were Site A: n=193 (7 women reported not ever having had sex) and Site B: n=298 (2 women were missing on age; 6 women reported not ever having had sex). All analyses were conducted using Stata version 15. Bivariate associations were assessed using t-tests and Chi-squared tests; a p value of 0.05 or less was considered statistically significant. A sum score representing ownership of four items (motorcycle,

car, radio, and television) served as a proxy for household wealth. Questions regarding sexually transmitted infections (STIs) were introduced into the Site B survey following input from Tibetan colleagues and health care providers involved with the Site A health fair.

Ethical consideration

Ethical clearance for data collection was provided by TUTMC. Participants were interviewed individually in a separate area from the health fair. The survey was explained to each woman and, due to limited literacy in this population, were asked to provide their oral consent.

Table 1 Sociodemographic characteristics of Tibetan Nomadic Women, 2014 and 2016

	2014 Site A	2016 Site B N = 298	
	n = 193		
Sociodemographic characteristics			
Mean age of respondents (range; median)	36.9 (21–50; 37) years	41.0 (18-55; 42) years	
Marital status			
Never married	5.2%	4.4%	
Married	82.4%	86.2%	
Divorced	8.8%	5.1%	
Widowed	3.6%	4.4%	
Educational attainment			
% reporting 0 years	80.2%	72.0%	
Average number of years (median) ^a	5.5 years (6)	4.3 years (4)	
Household wealth (asset score: 0–4)	1.9 (0-4; 0.8)	2.2 (0-4; 0.7)	
Enrollment in New Cooperative Medical System (NCMS), %	83.3%	83.9%	
Pregnancy and birth histories	Mean (range; median)		
Average number of pregnancies	3.0 (1–7; 3)	2.4 (0-8; 2)	
Average number of live births	2.7 (0–6; 3)	2.1 (0–6; 2)	
Age at first sex	17.1 (11–24; 17)	18.2 (7–27; 18)	
Age at first birth	21.0 (16–33; 20)	20.8 (13–37; 20)	
Ever had a miscarriage, %	16.1%	10%	
Ever had an abortion ^b	0.5%	14.5%	
Contraceptive use			
Report ever use of contraception	94.3%	83.5%	
Report current use of contraception	82.9% ^c	55.9% ^d	
IUD	73.8%	77.4% ^e	
Tubal ligation ^f	22.5%	8.8%	
Pill	1.3%	6.6%	
Condom	1.9%	5.8%	
Diaphragm	0.6%	_	
Injection	-	1.5%	

^a Among women with any education

^b Questions were asked differently between the two surveys. The 2014 survey asked women to report "How many times a pregnancy ended with an abortion?" The 2016 survey asked women to report each pregnancy first, then to report the outcome of each pregnancy

^c Survey did not ascertain current pregnancy status of women

^d N=250 who reported current contraceptive use. Omitted 5 women who were currently pregnant (n=245)

 $^{\rm e}\,$ 2 women who reported IUD and condom use were included in the IUD category

^f Two women who reported current use of injection and tubal ligation were recoded as having had a tubal ligation

Following the surveys, the data were maintained by the lead investigator at TUTMC in password-protected files and fully anonymized. The de-identified information was provided to investigators at University of California, Los Angeles (UCLA) for secondary data analysis. The UCLA Office of the Human Research Protection Program determined that analysis of these data did not meet the definition of human subjects research, so exempt this study from review.

Results

Table 1 displays the sociodemographic and reproductive characteristics of the two samples. The samples predominantly consist of women in older reproductive ages (Site A mean = 36.9 years; Site B mean = 41.0 years). At both sites, most women reported being currently married, and approximately 5% reported never being married and 10% reported being divorced or widowed. The vast majority of women reported no formal education (80% and 72% in Site A and Site B, respectively); of those women who reported any education, women reported 4–5 years on average. Households reported modest levels of household wealth and the vast majority were enrolled in the New Cooperative Medical Scheme, [15].

Women reported 2–3 lifetime pregnancies and births, on average, with women from the Site A sample reporting more pregnancies and births on average, as compared to the Site B sample (3.0 vs. 2.4 for pregnancies; 2.7 vs. 2.1 for live births, respectively). Ages at first sex and first birth were similar across the two sites, although women in Site A reported a larger delay between first sex and first birth (17.1 and 21.0 years, average delay of 3.2 years) as compared to women in Site B (18.2 and 20.8 years, average delay of 2.6 years). Differences in the questions regarding miscarriage and abortion across the two surveys prevent comparison across sites; however, 14.5% of women in the Site B survey and 0.5% of women in the Site A survey reported ever having an abortion.

Women reported high levels of ever using contraception (94% and 84% in Site A and Site B, respectively), and of current use (83% and 56%, respectively). Both populations indicate high current reliance on intrauterine devices (IUDs) and tubal ligation, as compared to other contraceptive methods. Reported use of the pill, condom, and injection were low, especially in Site A.

Data on awareness of STIs and possible STI-related symptoms were collected in the Site B survey only (Table 2). Just over half (58.6%) of the respondents reported they were aware that some infections and diseases may be transmitted sexually. In total, 76% of women reported ever experiencing any STI symptom, most commonly vaginal discharge (59.2%), itchiness (50.6%), and painful urination (20.6%). Twenty-one percent of women reported ever doing anything to prevent an STI.

We examined whether there were reported differences in the STI measures in the Site B survey between older women (40 years and above) and younger women (less than 40 years). Compared to older women, younger women were significantly more likely to report ever experiencing an STI symptom (p < 0.05) (Table 2). When looking at specific symptoms, experience of discharge and painful sex were more likely among younger versus older women.

Further examination of a potential risk factor—number of sexual partners—revealed that there was a marginally significant difference in the average number of lifetime sexual partners reported by women 40 years and above (2.2) versus women less than 40 years (1.9); however, significantly more women 40 years and above had only 1 lifetime partner, as compared to younger women (less than 40 years) (57.0% versus 43.1%, respectively, p value < 0.05). The average number of recent sexual

Table 2 Reported awareness of	f STIs and STI symptoms amo	ng Tibetan nomadic womer	n, Site B County	2016 /
-------------------------------	-----------------------------	--------------------------	------------------	--------

	Total	Women < 40 (n = 118)	Women 40+ (n = 180)
Awareness of STIs	59%	55%	61%
Ever experienced any STI symptom	76%	84%	71%*
Ever experienced			
Painful urination	21%	23%	19%
Itchiness	51%	52%	50%
Discharge	59%	75%	47%***
Painful sex	11%	14%	7%†
Genital warts	17%	19%	17%
Ever done anything to prevent STI	21%	26%	18%
Number of lifetime sexual partners	2.0 (1-15; 1.0)	2.2 (1–10; 2.0)	1.9 (1–15; 1.0) [†]
Number of sexual partners in last month	0.9 (0-2; 1)	1.0 (0–2; 1)	0.9 (0-1; 1)***

t-test of difference: [†]p < 0.1, *p < 0.05, **p < 0.01, ***p < 0.001

partners (within the last month) was significantly less among women 40 years and above (0.9 partners) than women less than 40 years old (1 partner) (p value < 0.001).

Discussion

This paper provides descriptive sociodemographic and reproductive health data collected from a hard-to-reach population—Tibetan nomads in Western China. Due to the sparse distribution of the population and their remote location, we implemented a novel and unique way of gathering data from women via community health fairs. This enabled both men and women from this underserved population to access health services, while also providing a venue for collecting basic health information.

Examination of the sociodemographic characteristics of these two samples indicate low education levels among participating women (72–80% reported no formal education). These levels are lower than those of women who participated in a separate study in a nearby agricultural area [26, 27], yet similar to educational levels documented in another study of nomadic Tibetan women [4]. Women's education is a critical component in improving women's status within societies and in achieving optimal health outcomes for women and their families [28, 29]. Improved opportunities for schooling, including for women and girls, are critical to achieving better health outcomes and well-being, while also maintaining Tibetan cultural identity and language [30].

Fertility levels reported among these samples (2.7 and 2.1 live births) likely approximate total fertility rates given that a large proportion of women from these samples were likely to have completed childbearing. (Additional analyses of Site B data did not indicate significant cohort differences in fertility between women less than 40 years versus women 40 years and above) Of note is that in both populations, sexual debut was reported to be at least two years prior to reported age at marriage and first birth, indicating that premarital sex is occurring and perhaps with the use of contraception. This finding is in contrast to some other Asian populations and aligns with past ethnographic research on the "loose association" between marriage and fertility in Tibetan societies ([31-33]—from Spoorenberg) and may suggest that young unmarried women are able to access contraception to delay pregnancy. It is also important to consider, however, the extent to which subfecundity or infertility may affect the ability to conceive (at any stage of the reproductive lifespan) given historical evidence of higher rates of primary and secondary infertility among Tibetan versus Han populations [34, 35].

The surveys did not collect data on types of contraception used throughout the lifecourse; however, women in the sample who reported current use of contraception predominantly relied on IUDs and tubal ligations. China has a history of high-reliance on IUDs and female sterilization [36]. While these methods are highly effective in preventing pregnancy and may be the preferred methods for this lower-fertility, remote population, it is also important that women (and men) have access to the full range of desired methods and high-quality contraceptive services [37]. We were not able to ascertain from the data the extent to which other contraceptive methods are used by younger women who wish to delay or space their pregnancies using short-term or reversible contraceptive methods.

Over three-quarters (76%) of the 2016 survey sample indicated that they had experienced at least one symptom that can be associated with STIs. It is important to note that while these symptoms are included in international surveys as a means of possible identification of STI infection, they are not highly specific to STIs and may reflect other infections or morbidities (e.g., urinary tract infection) [38]. It is also important to interpret these selfreported results cautiously—especially given the potential stigma and ongoing challenges with the validity of self-reported measures of sexual behavior across global settings [39].

Despite these measurement challenges; however, the significantly higher reporting of "ever" having an STI-related symptom among younger versus older women is notable which may reflect differential risk, as suggested by the marginally significant and higher mean number of lifetime sexual partners among younger women (2.2) as compared to older women (1.9). Alternately, it is possible that there may be different levels of comfort and reporting of number of sexual partners between younger and older women.

Although the community health fair provided an innovative approach to data collection among this marginalized population, there are challenges to this approach. Namely, we did not know how many women would attend the community health fair, nor did we know the characteristics of the women who would be able to or would be most willing to attend. The samples were older than expected, with mean ages of 37 (Site A) and 41 (Site B) at the two sites. Although the age of the sample allowed us to collect data on a range of reproductive experiences and history, in this relatively low-fertility population, for a proportion of the women in the sample, these reproductive experiences occurred nearly two decades prior.

The findings from this exploratory study should also be considered within the context in which the data were collected—although the health fair provided a venue for data collection among this marginalized population, data collected at the health fair selectively reflect the characteristics of women who chose to (and were able to) attend the health fair and who agreed to participate in the survey. Given the difficulty of reaching this marginalized population, this study provides initial insights on the health status and areas for continued investigation. Moreover, if we consider that that women who attended these health fairs may have had greater resources or capacity to attend such an event, it is possible that sociodemographic and health indicators (e.g., low levels of educational attainment and STI awareness) could be even lower within the broader population.

Further studies are needed to determine if the substantial environmental and structural changes in Western China may be altering risk exposure to STIs among ethnic minority populations. Rises in STI and HIV incidence have been documented in other settings marked by economic growth and expansion of transportation [40], and migration [41]. In addition to these broader contextual changes, there are aspects of the Tibetan lifestyle that are also changing-anecdotal data indicate that Tibetans in this region are increasingly employed in construction and road work projects that require them to move frequently, especially as other economic options (e.g., harvesting caterpillar fungus) become less accessible. In addition, our data collection effort was unfortunately limited to women and, thus, we are unable to comment on the potential change in risk and sexual behavior among Tibetan nomadic men.

Conclusions

Overall, findings from this study demonstrate the feasibility of collecting data via health fairs among this hardto-reach, remote, ethnic minority population. While timing of reproductive events was relatively stable across the age cohorts, high reporting of STI symptoms warrants further investigation to identify and address health conditions in this population of Tibetan nomadic women, especially amidst broader social and contextual changes that may affect the risk profiles of the Tibetan population.

Abbreviations

STI: Sexually transmitted infection; IUDs: Intrauterine devices.

Acknowledgements

We are grateful for support for this project from UCLA Bixby Center for Population and Reproductive Health, UCLA Asia Institute Title VI Joint East Asian Studies Center, UCLA Center for the Study of Women, Qinghai Province Thousand Talents Program (2018 Qing hai sheng "gao duan chuang xin ren cai qian ren ji hua"xiang mu 2018年青海省"高端创新人才千人计划"项目), TUTMC's Master Teacher Training Program (2018 nian qing hai da xue"ming shi pei yu ji hua"xiang mu 2018年青海大学"名师培育计划"项目), and State Key Laboratory of Tibetan Medicine (Qing hai sheng zang yao xin yao kai fa zhong dian shi yan shi xiang mu (2017-ZJ-Y15) 青海省藏药新药开发重点实验室项 目2017-ZJ-Y15).

Authors' contributions

JG completed data analysis and was the lead writer of the manuscript. CM assisted in analysis and writing. KG (Gangue Jianzan) oversaw data collection and provided input to manuscript. LT (Limao Cairang) assisted in data collection and provided input to manuscript. TN assisted in data collection and provided input to manuscript. LG (Li Xianjia) assisted in data collection and provided input to manuscript. All authors read and approved the final manuscript.

Funding

This project received funding from the UCLA Bixby Center for Population and Reproductive Health, UCLA Asia Institute Title VI Joint East Asian Studies Center, UCLA Center for the Study of Women, Qinghai Province Thousand Talents Program (2018 Qing hai sheng "gao duan chuang xin ren cai qian ren ji hua"xiang mu 2018年青海省"高端创新人才千人计划"项目), TUTMC's Master Teacher Training Program (2018 nian qing hai da xue "ming shi pei yu ji hua"xiang mu 2018年青海大学"名师培育计划"项目), and State Key Laboratory of Tibetan Medicine (Qing hai sheng zang yao xin yao kai fa zhong dian shi yan shi xiang mu (2017-ZJ-Y15) 青海省蕨药新药开发重点实验室项目 2017-ZJ-Y15).

Availability of data and materials

The datasets generated during and/or analysed during the current study are not publicly available due to the confidential and sensitive nature of the study population; however, these data may be made available from the corresponding author on reasonable request.

Ethics approval and consent to participate

TUTMC approved data collection. UCLA Office of the Human Research Protection Program deemed the analysis of secondary data exempt from review.

Consent for publication

Not applicable.

Competing interests

Not applicable.

Author details

¹ Fielding School of Public Health, University of California, 650 Charles E. Young Drive South, Los Angeles, CA 90095, USA. ² Tso-Ngon (Qinghai) University Tibetan Medical College, No. 251 Ningda Road, Xining 810000, Qinghai, People's Republic of China. ³ Department of Public Health, California State University, 800 North State College, Boulevard, KHS 131, Fullerton, USA.

Received: 20 August 2020 Accepted: 2 December 2020 Published online: 17 March 2021

References

- 1. Fischer AM. "Population invasion" versus urban exclusion in the Tibetan areas of Western China. Popul Dev Rev. 2008;34(4):631–62.
- Goodman DSG. The campaign to "Open up the west": national, provincial-level and local perspectives. China Q. 2004;178:317–34.
- UNDP C, Development Research Center of the State Council of China. Social innovation for inclusive human development. 2016.
- Wellhoner M, Lee AC, Deutsch K, Wiebenga M, Freytsis M, Drogha S, et al. Maternal and child health in Yushu, Qinghai province, China. Int J Equity Health. 2011;10:42.
- Yanqiu G, Ronsmans C, Lin A. Time trends and regional differences in maternal mortality in China from 2000 to 2005. Bull World Health Organ. 2009;87:913–20.
- Fang P, Dong S, Xiao J, Liu C, Feng X, Wang Y. Regional inequality in health and its determinants: evidence from China. Health Policy. 2010;94(1):14–25.
- Huang Y, Shallcross D, Pi L, Tian F, Pan J, Ronsmans C. Ethnicity and maternal and child health outcomes and service coverage in western China: a systematic review and meta-analysis. Lancet Glob Health. 2018;6(1):e39–56.

- Petousi N, Robbins PA. Human adaptation to the hypoxia of high altitude: the Tibetan paradigm from the pregenomic to the postgenomic era. J Appl Physiol (1985). 2014;116(7):875–84.
- Yang J, Jin ZB, Chen J, Huang XF, Li XM, Liang YB, et al. Genetic signatures of high-altitude adaptation in Tibetans. Proc Natl Acad Sci USA. 2017;114(16):4189–94.
- Foggin PM, Torrance ME, Dorje D, Xuri W, Marc Foggin J, Torrance J. Assessment of the health status and risk factors of Kham Tibetan pastoralists in the alpine grasslands of the Tibetan plateau. Soc Sci Med. 2006;63(9):2512–32.
- Wu Y, Zhou H, Wang Q, Cao M, Medina A, Rozelle S. Use of maternal health services among women in the ethnic rural areas of western China. BMC Health Serv Res. 2019;19(1):179.
- Song P, Kang C, Theodoratou E, Rowa-Dewar N, Liu X, An L. Barriers to hospital deliveries among ethnic minority women with religious beliefs in China: a descriptive study using interviews and survey data. Int J Environ Res Public Health. 2016;13(8):815.
- 13. Schrempf M. Re-production at stake: experiences of family planning and fertility among Amdo Tibetan Women. Asian Med. 2010;11:321–47.
- Moucheraud C, Gyal L, Gyaltsen K, Tsering L, Narasimhan S, Gipson J. Maternal health behaviors and outcomes in a Nomadic Tibetan population. Matern Child Health J. 2018;22(2):264–73.
- 15. You X, Kobayashi Y. The new cooperative medical scheme in China. Health Policy. 2009;91(1):1–9.
- Gao Y, Zhou H, Singh NS, Powell-Jackson T, Nash S, Yang M, et al. Progress and challenges in maternal health in western China: a Countdown to 2015 national case study. Lancet Glob Health. 2017;5(5):e523–36.
- Feng XL, Xu L, Guo Y, Ronsmans C. Socioeconomic inequalities in hospital births in China between 1988 and 2008. Bull World Health Organ. 2011;89(6):432–41.
- Liu X, Gao W, Yan H. Measuring and decomposing the inequality of maternal health services utilization in Western Rural China. BMC Health Serv Res. 2014;14(1):1.
- Goldstein MC, Jiao B, Beall CM, Tsering P. Fertility and family planning in rural Tibet. China J. 2002;47:19–39.
- Craig SR, Childs G, Beall CM. Closing the womb door: contraception use and fertility transition among culturally tibetan women in highland Nepal. Matern Child Health J. 2016;20(12):2437–50.
- Parish WL, Laumann EO, Cohen MS, Pan S, Zheng H, Hoffman I, et al. Population-based study of chlamydial infection in China: a hidden epidemic. JAMA. 2003;289(10):1265–73.
- Liao KJ, Zhang SK, Liu M, Wang QM, Liu J, Shen HP, et al. Seroepidemiology of syphilis infection among 2 million reproductive-age women in rural China: a population-based, cross-sectional study. Chin Med J (Engl). 2017;130(18):2198–204.
- Fu Z, He N, Duan S, Jiang Q, Ye R, Pu Y, et al. HIV infection, sexual behaviors, sexual networks, and drug use among rural residents in Yunnan Province, China. AIDS Behav. 2011;15(5):1017–25.
- Luo ZZ, Li W, Wu QH, Zhang L, Tian LS, Liu LL, et al. Population-based study of chlamydial and gonococcal infections among women in Shenzhen, China: Implications for programme planning. PLoS ONE. 2018;13(5):e0196516.

- ICF. DHS Model Questionnaire—Phase 6 (2008–2013) (English, French) The DHS Program website. https://dhsprogram.com/publications/publi cation-dhsq6-dhs-questionnaires-and-manuals.cfm.
- Gipson JD, Gyaltsen K, Gyal L, Kyi T, Hicks AL, Pebley AR. Tibetan women's perspectives and satisfaction with delivery care in a rural birth center. Int J Gynaecol Obstetr. 2015;129(3):244–7.
- 27. Gyaltsen K, Gipson JD, Gyal L, Kyi T, Hicks AL, Pebley AR. Maternal health care seeking by rural Tibetan women: characteristics of women delivering at a newly-constructed birth center in western China. BMC Pregnancy Childbirth. 2015;15:225.
- Ahmed S, Creanga AA, Gillespie DG, Tsui AO. Economic status, education and empowerment: implications for maternal health service utilization in developing countries. PLoS ONE. 2010;5(6):e11190.
- Gakidou E, Cowling K, Lozano R, Murray CJ. Increased educational attainment and its effect on child mortality in 175 countries between 1970 and 2009: a systematic analysis. Lancet. 2010;376(9745):959–74.
- 30. Yang MY. Learning to be safe citizens: state-run boarding schools and the dynamics of Tibetan identity. Citizensh Stud. 2017;21(7):824–41.
- Goldstein MC. New perspectives on Tibetan fertility and population decline. Am Ethnol. 1981;8(4):721–38.
- 32. Ma R. Marriage and spouse selection in Tibet. Dev Soc. 2001;30(1):79–117.
- 33. Childs G. Tibetan transitions: Historical and contemporary perspectives on fertility, family planning, and demographic change. Leiden: Brill; 2008.
- Spoorenberg T. Sixty years of change in Tibetan fertility: an assessment. Popul Stud (Camb). 2019;73(2):277–85.
- Liu J, Larsen U, Wyshak G. Prevalence of primary infertility in China: in-depth analysis of infertility differentials in three minority province/ autonomous regions. J Biosoc Sci. 2005;37(1):55–74.
- Bertrand JT, et al. Contraceptive method skew and shifts in method mix in low- and middle-income countries. Int Perspect Sex Reprod Health. 2014;40(3):144–53.
- Sullivan TM, et al. Skewed contraceptive method mix: why it happens, why it matters. J Biosoc Sci. 2006;38(4):501–21.
- Pettifor A, Walsh J, Wilkins V, Raghunathan P. How effective is syndromic management of STDs?: a review of current studies. Sex Transm Dis. 2000;27(7):371–85.
- Cleland J, Boerma JT, Carael M, Weir SS. Monitoring sexual behaviour in general populations: a synthesis of lessons of the past decade. Sex Transm Infect. 2004;80(Suppl 2):ii1–7.
- Botao C, Horth RZ, Frank H, Cummings B, Inguane C, Sathane I, et al. Prevalence of HIV and associated risk factors among long distance truck drivers in Inchope, Mozambique, 2012. AIDS Behav. 2016;20(4):811–20.
- El-Bassel N, Gilbert L, Shaw SA, Mergenova G, Terlikbayeva A, Primbetova S, et al. The silk road health project: how mobility and migration status influence HIV risks among male migrant workers in Central Asia. PLoS ONE. 2016;11(3):e0151278.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

